Author's response to reviews

**Title:** Protective effects of chebulic acid on alveolar epithelial damage induced by urban particulate matter

**Authors:**

Kyung-Won Lee (lee.kw129@gmail.com)

Mi-Hyun Nam (midnight1106@korea.ac.kr)

Hee-Ra Lee (zgmlfkz@naver.com)

Chung-Oui Hong (hco5979@korea.ac.kr)

Kwang-Won Lee (kwangwon@korea.ac.kr)

**Version:** 2 **Date:** 26 Jun 2017

**Author’s response to reviews:**

Editorial comments:

- We note that you have added an author to your manuscript during the review process. While we are able to accept changes to the authorship of your manuscript such as this, we do require approval from all authors for this change in order to proceed. We must therefore ask you to please complete the attached 'change of authorship' form, to obtain the signatures of all manuscript authors, and to then return this form to us.

Answer: We obtained the signatures of all manuscript authors and attached ‘Change of authorship’ form.

- We would ask you to please remove all supplementary files, except for those which contain supplementary data, from your submission. While we thank you for providing us with each of these, we do not consider it necessary for any of them to be published as additional files alongside your manuscript.

Answer: We removed all supplementary files, except for those which contain supplementary data, from our manuscript.